Author's response to reviews

Title: Understanding long-term sick leave in female white-collar workers with burnout and stress-related diagnoses: a qualitative study

Authors:

Helene Sandmark (helene.sandmark@oru.se)
Monica Renstig (monica.renstig@wbs.se)

Version: 4 Date: 13 April 2010

Author's response to reviews: see over
Covering letter April 13th 2010.

Dear Editor,

We have made the revisions of the manuscript that you required. We added some text in the sections “Authors contribution”, removed trackings and did some minor typographic changes in the text.

Kind regards,

Hélène Sandmark